Author's response to reviews

Title: Poor mental health and sexual risk behaviours in Uganda: A cross-sectional population-based study

Authors:

Patric Lundberg (patric.lundberg@ki.se)
Godfrey Rukundo (rukundogzari@yahoo.co.uk)
Schola Ashaba (scholaashaba@yahoo.com)
Anna Thorson (anna.thorson@ki.se)
Peter Allebeck (peter.allebeck@ki.se)
Per-Olof Östergren (Per-Olof.Ostergren@med.lu.se)
Elizabeth Cantor-Graae (elizabeth.cantor-graae@med.lu.se)

Version: 3 Date: 10 February 2011

Author's response to reviews: see over
Dear Sirs,

Please find below the changes that we have made on the final manuscript. Unfortunately, due to a mistake, the first two changes are not visible as tracked changes.

1. Page 14, Interpretation, second sentence: “..this association..” has been replaced by “these associations”.
2. A number of references lacked year of publication, this has been corrected.
3. Page 17, Methodological considerations, paragraph starting with “Follow-up visits were not performed..”. The second sentence in this paragraph has been slightly modified.
4. Minor corrections have been made in the Tables.

Yours sincerely,

Patric Lundberg